

**RESEARCH ARTICLE** 

# Evolutionary Trade-Offs Underlie the Multifaceted Virulence of *Staphylococcus aureus*

Maisem Laabei<sup>1</sup>, Anne-Catrin Uhlemann<sup>2</sup>, Franklin D. Lowy<sup>2</sup>, Eloise D. Austin<sup>2</sup>, Maho Yokoyama<sup>1</sup>, Khadija Ouadi<sup>1</sup>, Edward Feil<sup>1</sup>, Harry A. Thorpe<sup>1</sup>, Barnabas Williams<sup>1</sup>, Mark Perkins<sup>1</sup>, Sharon J. Peacock<sup>3</sup>, Stephen R. Clarke<sup>4</sup>, Janina Dordel<sup>5</sup>, Matthew Holden<sup>5,6</sup>, Antonina A. Votintseva<sup>7</sup>, Rory Bowden<sup>8</sup>, Derrick W. Crook<sup>7</sup>, Bernadette C. Young<sup>7</sup>, Daniel J. Wilson<sup>7,8°</sup>, Mario Recker<sup>9°</sup>, Ruth C. Massey<sup>1°\*</sup>

1 Department of Biology and Biochemistry, University of Bath, Bath, United Kingdom, 2 Department of Medicine, Columbia University, New York, New York, United States of America, 3 Department of Medicine, University of Cambridge, Cambridge, United Kingdom, 4 School of Pharmacy and Pharmaceutical Sciences, Cardiff University, Cardiff, United Kingdom, 5 Pathogen Genomics, Wellcome Trust Sanger Institute, Wellcome Trust Genome Campus, Hinxton, Cambridge, United Kingdom, 6 School of Medicine, Medical & Biological Sciences, North Haugh, St Andrews, Fife, United Kingdom, 7 Nuffield Dept. of Medicine, University of Oxford, John Radcliffe Hospital, Oxford, United Kingdom, 8 Wellcome Trust Centre for Human Genetics, Oxford, United Kingdom, 9 Centre for Mathematics and the Environment, University of Exeter, Penryn Campus, Penryn, United Kingdom

These authors contributed equally to this work.
r.c.massey@bath.ac.uk

## Abstract

Bacterial virulence is a multifaceted trait where the interactions between pathogen and host factors affect the severity and outcome of the infection. Toxin secretion is central to the biology of many bacterial pathogens and is widely accepted as playing a crucial role in disease pathology. To understand the relationship between toxicity and bacterial virulence in greater depth, we studied two sequenced collections of the major human pathogen Staphylococcus aureus and found an unexpected inverse correlation between bacterial toxicity and disease severity. By applying a functional genomics approach, we identified several novel toxicityaffecting loci responsible for the wide range in toxic phenotypes observed within these collections. To understand the apparent higher propensity of low toxicity isolates to cause bacteraemia, we performed several functional assays, and our findings suggest that within-host fitness differences between high- and low-toxicity isolates in human serum is a contributing factor. As invasive infections, such as bacteraemia, limit the opportunities for onward transmission, highly toxic strains could gain an additional between-host fitness advantage, potentially contributing to the maintenance of toxicity at the population level. Our results clearly demonstrate how evolutionary trade-offs between toxicity, relative fitness, and transmissibility are critical for understanding the multifaceted nature of bacterial virulence.

## Author Summary

Global efforts to counter the growing problem of antibiotic resistance and develop alternative treatment strategies rely on a fuller understanding of when and why opportunistic



## 

**Citation:** Laabei M, Uhlemann A-C, Lowy FD, Austin ED, Yokoyama M, Ouadi K, et al. (2015) Evolutionary Trade-Offs Underlie the Multi-faceted Virulence of *Staphylococcus aureus*. PLoS Biol 13(9): e1002229. doi:10.1371/journal.pbio.1002229

Academic Editor: David S. Schneider, Stanford University, UNITED STATES

Received: May 21, 2015

Accepted: July 20, 2015

Published: September 2, 2015

**Copyright:** © 2015 Laabei et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

Funding: ML is funded on an MRC Confidence in Concepts award. MR is funded by a Royal Society Research Fellowship. The single-patient study was supported by the Oxford NIHR Biomedical Research Centre and the UKCRC Modernising Medical Microbiology Consortium, the latter funded under the UKCRC Translational Infection Research Initiative supported by the Medical Research Council, the Biotechnology and Biological Sciences Research Council and the National Institute for Health Research on behalf of the UK Department of Health (Grant



G0800778) and the Wellcome Trust (Grant 087646/Z/ 08/Z). We acknowledge the support of Wellcome Trust core funding (Grant 090532/Z/09/Z). DWC is an NIHR Senior Investigator. BCY is a Wellcome Trust Clinical Research Training Fellow (Grant 101611/Z/ 13/Z). DJW is a Sir Henry Dale Fellow, jointly funded by the Wellcome Trust and the Royal Society (Grant 101237/Z/13/Z). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

Abbreviations: BHI, brain-heart infusion broth; ENC, early nose culture; GWAS, genome-wide association study; LBC, late blood culture; LNC, late nose culture; NET, neutrophil extracellular trap; PSM, phenol soluble modulins; PVL, Panton-Valentine-leukocidin; SSTI, skin and soft tissue infection. pathogens cause disease. Recent advances in DNA sequencing technologies have opened up new opportunities to study infectious organisms, yet identifying the genetic variants that explain differences in disease remains challenging. Here we aimed to understand the complex relationship between toxicity—a known risk factor for disease in many bacteria and infection severity for the major human pathogen *S. aureus*. Against expectations, we found that the bacteria that caused the most severe disease were the least toxic strains. We were able to determine the mutations responsible for the differences in toxicity and identified a number of novel toxicity-affecting genes. We further discovered that bacterial fitness in human serum could explain the unexpected association of low-toxicity isolates with severe, invasive disease. Invasive *S. aureus* infections are usually considered a dead end for these bacteria, as these infections are rarely transmitted to another person. Here we show using a simple mathematical model that this might in fact favour transmission of highly toxic bacteria on a population level and thus contribute to their global success. Our work therefore highlights the complexity of bacterial infection and should aid in devising new treatment and control strategies against this important pathogen.

#### Introduction

The development of effective, long-term control strategies against microbial pathogens crucially relies on a thorough understanding of the many factors that contribute to the evolution and maintenance of enhanced virulence. Bacterial toxins are well established as playing a key role in virulence [1-4]. They release nutrients for bacterial growth and facilitate intra- and interhost transmission by destroying local tissue and subverting host immune processes [5-7]. This has led to the general presumption that elevated toxicity is positively associated with enhanced disease severity in bacterial infections [5,8,9]. However, the expression of toxins is readily switched off in vitro in response to the selection imposed by the energetically costly nature of their production [10,11]. Observational studies in *S. aureus* suggest that this can also occur in vivo, indicating that the relationship between toxicity and disease severity is more complex than initially appreciated [12-17].

*S. aureus* is a major human pathogen and a global healthcare issue. It is considered opportunistic as it asymptomatically colonises its host but can occasionally cause diseases that range in severity from relatively minor skin and soft tissue infections (SSTI) to life-threatening cases of pneumonia and bacteraemia [9]. Toxicity has been accepted as playing a key role in the success of lineages such as USA300 [18] and ST93 [19], in which it has been suggested to increase transmissibility. Toxicity is also widely accepted as playing a significant role in the virulence of *S. aureus*, where many studies comparing high- and low-toxicity isolates in animal models of sepsis show that highly toxic isolates cause more severe disease symptoms [20–24]. However, it has recently been shown that *S. aureus* isolates from humans with invasive diseases, such as bacteraemia and pneumonia, are often impaired in their ability to express toxins (often referred to as Agr dysfunction) [12,13]. A strength of the collections of isolates considered in these studies is that they represent all of the isolates presenting in a given geographical region over a defined period of time. Unfortunately, this type of sampling brings with it a limitation, as such collections consequently also contain a wide range of genetic backgrounds and antibiotic resistance profiles that might confound any potential associations between toxicity and disease.

In this study, we applied a robust functional genomics approach to two collections of *S. aureus* isolates, with significant depth and breadth, where we have controlled for genetic background by sampling within specific clinically important lineages, including USA300. Our

approach not only allowed us to make observations that challenge our understanding of the role of toxicity in the establishment of severe, invasive disease but also to identify the genetic polymorphisms involved. By using a combination of functional assays, we further identified bacterial fitness in human serum as an important factor that could limit the ability of highly toxic isolates to cause bacteraemia and would explain the observed negative relationship between toxicity and disease severity. The power of genomics to study past events is clear, and here we demonstrate its potential to also help us understand fundamental aspects of bacterial pathogenicity and their role in invasive disease.

#### Results

To determine the relationship between toxicity and disease for *S. aureus*, we performed a detailed investigation of two large collections of fully sequenced clinical isolates representing significant breadth and depth of sampling (<u>S1 Table</u>). The first collection (ST15, PVL-, MSSA) was isolated from a single patient who progressed from asymptomatic carriage to bacteraemia over a 15-mo period [<u>25</u>]. The second, a collection of 134 clinical isolates, all corresponding to a single clone (USA300, ST8, PVL+, SCCmec type IV MRSA) and isolated from the nose or skin of healthy volunteers (carriage), from SSTI or from bacteraemic patients [<u>26</u>].

#### Reduced Toxicity Associates with Invasive Disease

The single-patient collection included serial asymptomatic nasal carriage isolates over a 12-mo period, as well as bloodstream isolates after bacteraemia had developed at month 15 [25]. At each time point, 12 individual colonies were isolated from the primary plates for each swab. These isolates all belong to the ST15 lineage and contain genes for 12 of the 13 known cytolytic toxins secreted by S. aureus (i.e., alpha, beta, gamma, delta, LukAB, LukED, PSM $\alpha$ 1,  $\alpha$ 2,  $\alpha$ 3, 04, β1, β2, but not LukSF [a.k.a. Panton-Valentine-leukocidin or PVL]). To quantify the gross cytolytic activity of each isolate, we used an immortalised T/B hybridoma cell line (T2), which is susceptible to 10 of the 12 cytolytic toxins present in this collection (not LukAB or LukED) (S1 Fig). Despite some sequence variability across the 12 isolates from each time point and between the different time points [25], no diversity in toxicity was observed for the early nasal culture isolates (Fig 1A). At month 12, however, there was a significant drop in toxicity for all 12 nasal carriage colonies, shortly after which the study participant developed bacteraemia (month 15). The bacteria isolated from the patient's bloodstream also showed significantly reduced toxicity compared to those from the earlier time points (Fig 1A), demonstrating an apparent inverse correlation between toxicity and disease, albeit in a sample size of only one patient.

To ensure this effect was not a consequence of the lack of sensitivity of the T2 cell line to LukAB and ED, we quantified the ability of a subset of six high- and six low-toxicity isolates (as determined using the T2 cells) to lyse freshly harvested human neutrophils, which are sensitive to LukAB and LukED. We found that the isolates that were unable to lyse the T2 cells were also unable to lyse the neutrophils (<u>S1 Fig</u>), suggesting that LukAB and LukED were not expressed by these low-toxicity isolates, and that the T2 cell line provides a robust measure of gross cytolytic activity for this collection of isolates.

Having demonstrated a negative correlation between toxicity and disease for a single patient, we sought to extend this to multiple patients by focussing on a collection of 134 USA300 MRSA isolates. As the USA300 lineage is known to contain the genes encoding the LukSF (i.e., PVL) toxin we included a human monocytic cell line, THP-1, which is sensitive to the action of this toxin [27]. In this set of isolates, we again found that the bacteraemic isolates were significantly less toxic than either the carriage or the SSTI groups (Fig 1B and 1C,



Fig 1. Low levels of toxicity are associated with invasive S. aureus infections across two collections of clinical isolates. a: Individual isolates from seven time points were assayed for toxicity and their genome sequenced. Early nose cultures (ENCs) from months 1, 2, 4, 6, and 8 were all highly toxic. Late nose culture (LNC) isolates at month 12 and bloodstream isolates following bacteraemia (late blood culture, LBC) were all significantly less toxic than the ENC isolates. b and c: The toxicity of 134 USA300 isolates were assayed, and those from bacteraemia were on average significantly less toxic (measured as percent T2 and THP1 cell

death (b and c, respectively)) than from carriage or SSTI (p < 0.01 ANOVA). The medians are presented as horizontal bars, with the boxes and whiskers showing the 1st and 3rd quartile and interquartile ranges. To access this data, see <u>S1 Data</u>.

doi:10.1371/journal.pbio.1002229.g001

ANOVA: p < 0.01 for both T2 cells and THP-1 cells). With these two collections of isolates, representing both broad sampling across patients and deep sampling within a patient, we demonstrate a significant negative association between toxicity and disease severity.

#### Polymorphisms in a Range of Loci Affect the Toxicity of S. aureus

The single-patient collection contained only a small number of genetic differences between the isolates, and as such we were readily able to identify the genetic basis of the change in toxicity. We found that the late nasal carriage and blood culture isolates had a premature stop-codon in an *araC*-like transcriptional regulator gene *rsp*, which encodes a protein that has previously been shown to regulate biofilm formation by this pathogen [28]. Using a transposon insertion in the *rsp* gene (NE1304) from the Nebraska library [29], we found a significant reduction in toxicity compared to the wild-type strain (Fig 2), thus verifying this gene's toxicity-regulating function. Interestingly, mutating the *rsp* gene with a transposon in the USA300 (ST8) background (Fig 1A), presumably as a consequence of differences in the genetic background of these bacterial clones.

To identify the genetic polymorphisms responsible for the changes in toxicity of the USA300 isolates, we adopted a functional genomics approach using genome-wide association studies (GWAS) to identify candidate polymorphisms associated with toxicity (S2 Table). As GWAS is prone to high false positive rates, we sought to verify the effect of the associated loci using transposon insertions in each locus that was available from the Nebraska library (the full list of significantly associated polymorphisms can be found in <u>S2 Table</u>). We identified five novel toxicity-affecting loci (Fig 2): ftsK, clpC, sucD, rpsA, plus a hypothetical gene with no known activity or homology to other proteins (SAUSA300\_0750). Based on amino acid homology, the FtsK protein is believed to be a DNA translocase, where polymorphisms have also been associated with changes in the toxicity of another globally successful MRSA clone, ST239 [30]. FtsK also shares significant structural similarities to transporters, suggesting that this protein may be directly involved in the secretion of toxins from the bacterial cell, rather than DNA translocation. The ClpC protein is a chaperone and has been shown previously to affect the expression of a large number of proteins, including several regulators of toxin expression [31]. The role of the proteins encoded by sucD (succinyl-CoA synthetase subunit alpha) and rpsA (30S ribosomal protein S1) in toxin expression is less clear, although any changes to the metabolism of a cell are likely to have significant downstream effects on gene expression [32]. Further work is underway to elucidate the molecular detail of how these proteins affect toxicity.

#### Explaining the Association between Toxicity and Infection Severity

A) Underlying health of patient. It is possible that the less toxic isolates are more often associated with bacteraemia because of the presence of comorbidities associated with an impaired immune system that might result in failure to control less toxic organisms. Although we cannot fully discount this, when we examined the relatedness of these isolates by mapping the isolate source and level of toxicity onto a maximum likelihood tree based on their genome sequence (Fig 3), despite some clustering of the bacteraemic isolates, they are distributed across the tree and are often closely related to highly toxic carriage or SSTI isolates. This suggests that



Fig 2. The identification of six novel toxicity-affecting loci. Transposon mutagenesis was used to functionally verify the role of loci associated with toxicity by genome-wide association study (GWAS; PLINK). Mean of six replicates are presented, error bars represent the 95% confidence intervals. The mutation in *agrB* is provided as a negative control. To access this data, see <u>S2 Data</u>.

doi:10.1371/journal.pbio.1002229.g002

PLOS

BIOLOGY

a susceptible patient would be just as or more likely to be exposed to a highly toxic as to a low toxicity isolate. A retrospective chart review (summarised in <u>S3 Table</u>) also found no associations between the presence of suspected portals of entry, types of suspected portals of entry, or length of bacteraemia with the toxicity of the isolates. While the health of the patient is undoubtedly a factor in their susceptibility to the development of bacteraemia, where the damage-response framework is clearly implicated [<u>33</u>], it does not explain why patients are more likely to develop invasive diseases with the less toxic isolates.

**B)** Cell-invasive capacity. The regulation of toxin secretion by *S. aureus* has been well studied under laboratory conditions [34,35], where they are known to phenotypically switch between their ability to attach to and invade human cells and their ability to secrete toxins. We therefore hypothesised that the more toxic isolates might be impaired in their ability to invade human cells, causing them to be less able to gain entry to the bloodstream. To test this hypothesis, we quantified the ability of 10 high- and 10 low-toxicity isolates to invade a human-derived endothelial cells line (EA.hy926) and found no differences in cell-invasive capacity between high- and low-toxicity isolates (Fig 4B).

**C)** The release of neutrophil extracellular traps (NETs). Another feature of the ability of *S. aureus* to cause disease is their immune evasive capacity. Neutrophils are of particular importance in immunity to *S. aureus* infection, and of the many roles they play, their ability to trap bacteria in NETs has been shown to be triggered by cytolytic toxins [36]. We therefore





**Fig 3. Toxicity levels and disease origin distributed across the USA300 collection.** The clinical origin (carriage = C, SSTI = S, or blood = B) and the toxicity (low = green, medium = orange or high = red) of 134 USA300 isolates are mapped onto a maximum likelihood tree based on their whole genome sequence.

doi:10.1371/journal.pbio.1002229.g003

hypothesised that neutrophils in the bloodstream might be more efficient at trapping the more toxic isolates, impeding their ability to establish an infection in the bloodstream. However, when we compared the ability of 10 high- and 10 low-toxicity isolates to induce neutrophil net formation, while we observed significant differences between individual isolates, this was unrelated to their toxicity (Fig 4C).

**D) Protease activity.** As the secretion of proteases by *S. aureus* also causes significant tissue damage, we hypothesised that the more toxic isolates might be less proteolytic, and this could impede their ability to cause bacteraemia. To test this, we assayed the protease activity of 10 high- and 10 low-toxicity isolates; however, no significant difference in protease activity between these collections of isolates was observed (Fig 4D)

**E)** Antimicrobial peptide resistance. A fundamental aspect of host innate immunity that provides protection from bacterial infection is the secretion of antimicrobial peptides. We hypothesised that the more toxic isolates may be less resistant to the effect of these, which





**Fig 4. Explaining the association between toxicity and infection severity.** A range of virulence assays were performed on a subset of 10 high- and 10 low-toxicity isolates from the USA300 collection to identify the biological feature that explains the observed inverse correlation between toxicity and disease severity. (A) The toxicity of the subset of isolates is provided for comparison. (B) The ability of the bacteria to invade EA.hy926 endothelial cells. (C) The

ability of the bacteria to induce the release of neutrophil extracellular traps (NETs). (D) The release of proteases by the bacteria. (E) Resistance of the bacteria to human neutrophil defensin-1. (F) The ability of the bacteria to form biofilms. The *p*-value for the comparison between the high and low toxicity isolates was greater than 0.05 for all traits with the exception of toxicity. The medians are presented as horizontal bars, with the boxes and whiskers showing the 1st and 3rd quartile and interquartile ranges. To access this data see <u>S3 Data</u>.

doi:10.1371/journal.pbio.1002229.g004

would impede their ability to cause bacteraemia. To test this, we measured the percentage survival of a subset of 10 high- and 10 low-toxicity isolates following incubation with 5  $\mu$ g/mL human neutrophil defensin-1 (hNP-1), but no significant difference between these collections of isolates was observed (Fig 4E)

**F) Biofilm formation.** As biofilm formation by *S. aureus* is believed to contribute to their ability to colonise both biotic and abiotic surfaces, we hypothesised that the high toxicity isolates may be less able to form biofilm, impeding their ability to cause bacteraemia. To test this, we quantified the ability of 10 high- and 10 low-toxicity isolates to form biofilms; however, no difference between the high and low toxicity isolates was observed (Fig 4F).

G) Relative fitness in human serum. The bloodstream is a highly protected niche. With almost all elements of host immunity present, it acts as a severe bottleneck for the bacteria, as evidenced by the lack of genetic diversity amongst the isolates from bacteraemic patients (we found no genetic diversity amongst them, despite the 12 isolates having been sampled at two time points and collected in three bottles). Serum contains many antimicrobial peptides and has been shown in two previous studies to have the effect of both increasing the expression of toxins [37,38] and neutralising their activity simultaneously [39]. As toxin expression is widely accepted as being an energetically costly activity, we hypothesised that the fitness costs associated with the up-regulation of toxin expression in blood could be sufficient to reduce the probability of the highly toxic isolates from getting through the bottleneck and establishing a blood stream infection. We therefore compared the growth dynamics and relative fitness of six highand six low-toxicity isolates from the single-patient collection and 10 high- and 10 low-toxicity isolates from the USA300 collection in brain-heart infusion broth (BHI) and in BHI supplemented with 5% human serum. Although 5% serum had significant antibacterial activity on these isolates (S2 Fig and S6 Data), we observed no difference in growth dynamics of the high and low toxicity isolates. However, when we measured the relative fitness of the isolates, which is more sensitive to growth defects, we found that in BHI the high-toxicity isolates from both collections were slightly less fit than the low-toxicity isolates, although not statistically significant (p = 0.19 and 0.28 respectively, Fig 5A and 5B). However, when serum was added, which has the effect of increasing toxin expression further, the difference in relative fitness between the high- and low-toxicity isolates was significantly increased for both collections (p = 0.003) and 0.02 respectively, Fig 5A and 5B).

To support our hypothesis that the effect on fitness is due to levels of toxin expression, we ensured that the low-toxicity isolates all had SNPs in a wide range of loci (including *rsp*, *ftsK*, *clpC*, *sucD*, *rpsA*, and SAUSA300\_0750), and further included an Agr wild type and mutant isogenic pair ( $\underline{Fig 5A-5C}$ ). Regardless of the locus affected, once toxicity was lowered, relative fitness was increased upon exposure to serum.

#### The Observed Association between Low Toxicity and Bacteraemia May Contribute to the Maintenance of High Toxicity at a Population Level

With mortality rates as high as 20% for *S. aureus* bacteraemia [40], understanding how these types of infections develop is of significant clinical importance. However, as the bacteria rarely transmit beyond this point, they can be effectively thought of as transmission dead ends. We therefore sought to explore the possible evolutionary effect of the observed association of low-





**Fig 5.** The highly toxic isolates are less fit than the low-toxicity isolates in the presence of human serum. We quantified the relative fitness of six high- and six low-toxicity isolates from the single patient collection (A), 10 high- and 10 low-toxicity isolates from the USA300 collection (B) and an Agr wild type and mutant isogenic pair (C) in BHI with and without 5% (vol/vol) human serum. The medians are presented as horizontal bars, with the boxes and whiskers showing the 1st and 3rd quartile and interquartile ranges. *p*-values are indicated in the text. To access this data, see S4 Data.

doi:10.1371/journal.pbio.1002229.g005

toxicity isolates with invasive infections using a simple mathematical model (see <u>Methods</u>). We considered two competing strains of *S. aureus* that differed in their level of toxicity, where higher toxicity was assumed to be positively correlated with transmissibility and progression from carriage to SSTI, based on the fact that we rarely find low-toxicity isolates amongst our carriage or SSTI populations, and that low-toxicity mutants have been rarely found to transmit amongst healthy populations [17]. But higher toxicity also resulted in faster treatment. With this, we examined what effect the strains' relative propensity to cause bacteraemia ( $\sigma$ ) had on their competitive fitness.

Assuming no differences ( $\sigma_l = \sigma_h$ , Fig 6A and 6B and S3 Fig), we find that the higher clearance rate of the more toxic strain offsets its transmission advantage, leading to its exclusion and dominance of the less toxic strain (shown separately for carriage and SSTI and bacteraemia in Fig 6A and 6B, respectively). In contrast, considering the clinically observed negative association of toxicity with bacteraemia ( $\sigma_l > \sigma_h$ ) results in the more toxic strain gaining a competitive advantage at the population level (carriage and SSTI, Fig 6C, S3 Fig and S4 Fig), whereas the less toxic strain maintains its elevated frequency during bacteraemia (Fig 6D, S3 Fig and S4 Fig), in line with clinical observations. This suggests that the reduced opportunity for transmission due to bacteraemia could partly compensate for the toxicity-driven trade-off between transmissibility and treatment rates and thus contribute to the maintenance and circulation of highly toxic strains within the population.

#### Discussion

For microbial pathogens, many factors contribute to their success, but for an opportunistic pathogen that can either reside asymptomatically or cause symptomatic infections ranging from superficial to life-threatening invasive disease, the definition of success becomes increasingly complex. The damage-response framework outlines the necessary holistic approach we need to take when considering this, where both the level of virulence expressed by the pathogen and the response of the host is critical to the clinical outcome [33]. For *S. aureus*, the vast majority of infections resolve without clinical intervention, and severe infections (e.g., bacteraemia or pneumonia) are generally limited to those with compromised health [9]. With many genes encoding cytolytic toxins, and highly toxic clones disseminating worldwide, it is understandable that toxicity and virulence have long been considered directly linked and key to its success. However, here we show this relationship to be quite complex.

The boom in genomic sequence data for microbial pathogens has allowed us to study past genetic events in great detail, tracing epidemics and studying how bacterial genomes evolve [25,41-44]. It is, however, only recently that we have been able to successfully use such genomic data to study the behaviour of pathogens and use functional genomics approaches to understand why and how specific traits evolve [30,45-49]. By studying two large collections of isolates, we demonstrate that bacteraemic isolates are significantly less toxic than those isolated from carriage or from SSTIs. With the genome sequence of each isolate available to us, we were able to identify the polymorphisms responsible for the observed changes in toxicity and in doing so have identified six novel toxicity-affecting loci for this pathogen. A molecular



**Fig 6. Evolution towards increased levels of virulence.** Two competing MRSA strains with different levels of toxicity are considered, where higher toxicity is assumed to increase the transition rate from carriage to infection (SSTI) but also increases treatment-induced clearance of infections. At baseline, we assume that both strains have equal propensities to cause severe bacteraemia ( $\sigma_l = \sigma_h$ ), which causes the more toxic strain (red) to be outcompeted due to its faster, virulence-induced clearance rates and leads to the dominance of the less toxic strain (green) both at carriage + SSTI (**a**) and at bacteraemic stage (**b**). By considering a negative correlation between toxicity and bacteraemia ( $\sigma_l > \sigma_h$ ), the more toxic strain gains a competitive advantage, leading to higher frequencies at carriage and SSTI stages (**c**). However, the higher propensity of the less toxic strain, in this case maintained in the population through mutational down-regulation from highly toxic strains, maintains its high frequency at the bacteraemic stage (**d**). See <u>Methods</u> for parameter values.

doi:10.1371/journal.pbio.1002229.g006

BIOLOGY

**PLOS** 

dissection of each locus is currently underway to determine how they affect this trait, but this work clearly demonstrates the power of functional genomics for studying bacteria.

To understand why low toxicity isolates have a higher propensity to cause bacteraemia, we developed and tested several hypotheses. While the health of a patient is a feature in their susceptibility to bacteraemia, we were unable to find evidence to suggest that this would increase their propensity to develop bacteraemia with a low rather than a highly toxic isolate. Furthermore, we found no evidence suggesting that either cell-invasiveness, NET formation, protease activity, antimicrobial peptide resistance, or biofilm formation play a role. Instead, we found that the presence of serum, which simultaneously increases toxin expression while neutralising their activity, reduces the relative fitness of the highly toxic isolates. Given the extreme bottleneck that the establishment of a bloodstream infection represents to bacteria, we believe this explains our finding that low-toxicity isolates are a more common cause of this type of invasive

disease. By use of a mathematical model, we also showed that this increased propensity for low-toxicity isolates to cause bacteraemia, alongside the dead end nature of such infections, could potentially contribute to the maintenance of high levels of toxicity at a population level, as evidenced by the global prevalence of highly toxic clones, such as USA300 and ST93.

To understand the relative efficiency of GWAS to identify novel toxicity-affecting loci, we need to compare it to a similar experiment that used a random approach. Fey et al., who created the transposon library used here, screened their 1952 Tn mutants for haemolytic activity using rabbit blood agar plates and identified 71 mutants with a change in this phenotype [29]. Compare this hit rate of 3.6% with ours of 6.7% and the relative efficiency of this GWAS approach is apparent. This efficiency is, however, significantly affected by the size and relatedness of the isolates within the collection; as in a previous GWAS study, we used a more closely related collection of isolates and had a hit rate of 30% [30]. Such issues need to be considered when designing GWAS experiments for bacteria.

This work highlights a clear limitation to existing animal models of infection for pathogens like *S. aureus*. Studies published by us and many other groups have shown that in sepsis models, highly toxic isolates cause a more severe infection in mice than less toxic isolates [20–24], which would seem contrary to our findings here in humans. However, a major difference between these two systems needs to be considered. For humans, studies have demonstrated that bacteraemia represents an extreme bottleneck, with only a small number of cells being sufficient to cause disease [25,50]. For mice, on the other hand,  $10^7-10^8$  colony-forming units need to be injected directly into the tail vein to get reproducible infections, which overrides the rigours bacteria need to go through to cause bacteraemia naturally in human. Furthermore, subtle differences in relative fitness, which we suggest contribute to our clinical observations, would not be evident in a mouse overwhelmed with the introduction of so many bacteria cells directly into its bloodstream. We would therefore urge caution when interpreting the results from such experiments in relation to human disease.

It is interesting to note that no difference in toxicity was observed between the carriage isolates and those causing SSTIs, and that, on average, both groups were highly toxic. Many animal models have been used to demonstrate that the size of a cutaneous lesion is greatly affected by the levels of toxins secreted by the infecting organism [5]. While we have no clinical data to compare toxicity levels and lesion size for these isolates, that we rarely find low-toxicity isolates causing SSTIs supports the findings from animal models that toxicity correlates with the ability to cause such lesions. It is also intriguing to consider what selective forces maintain such a high level of toxicity at a population level. For symptomatic transmission, the toxin-induced production of highly transmissible pus has obvious benefits. However for asymptomatic transmission, this is less obvious. Low toxicity isolates can survive in the nose, as indicated by the single patient collection where the low toxicity isolates survived there for 3 mo before causing bacteraemia. However, perhaps over a longer term, the ability of highly toxic bacteria to resist the effects of nasal-associated immunity by killing host immune cells, and the use of some toxins (e.g., the phenol soluble modulins, PSMs) to interfere with competing genera of bacteria, is key to its maintained selection.

For *S. aureus*, an opportunistic pathogen, it is clear that virulence is multifaceted. On the one hand, the prevalence of highly toxic clones globally and the role of specific toxins in causing highly transmissible pus-filled SSTI lesions suggests that toxins offer a selective advantage. On the other hand, it appears that offsetting toxicity for short-term enhanced fitness is associated with increased virulence, which may paradoxically select for the maintenance of higher levels of toxicity at a population level. Although at a superficial level these seem contradictory, it is clear that they are critical aspects of this pathogen's success. With the movement of genome sequencing into routine clinical practice and the drive towards personalised medicine,

we need to define these complex interactions and bring the biology of the pathogen into greater consideration in clinical settings.

## **Materials and Methods**

#### **Ethics Statement**

Whole blood was obtained from healthy volunteers; ethical permission for all donations was obtained from local research ethics committee (School of Pharmacy & Pharmaceutical Sciences Ethics Committee) and all participants gave written informed consent.

## **Bacterial Strains and Growth Conditions**

A list of the bacterial strains used can be found in <u>S1 Table</u>. For the toxicity assays, the *S. aureus* isolates were grown overnight in 5 mL of Tryptic-Soy Broth (TSB) in a 30 mL glass tube. This overnight culture was used to inoculate the toxicity-assay cultures at 1:1,000 dilution in fresh TSB and then grown for 18 h at 37°C in air with shaking (180 rpm). For transposon mutants, erythromycin (5  $\mu$ g/mL) was included in the growth medium. The toxin-containing supernatant for each isolate was harvested by centrifugation.

## T2 and THP-1 Toxicity Assay

Immortalised human T2 cells and monocyte macrophage THP-1 cell lines were used as described previously [51]. The THP-1 cell line was included for the USA300 collection as it is susceptible to the PVL [27]. Briefly, both cell lines were grown in individual 30 mL suspensions of RPMI-1640, supplemented with 10% heat-inactivated fetal bovine serum (FBS), 1 µM L-glutamine, 200 units/mL penicillin, and 0.1 mg/mL streptomycin at 37°C in a humidified incubator with 5% CO<sub>2</sub>. Cells were routinely viewed microscopically every 48-60 h and harvested by centrifugation at 1,000 rpm for 10 min at room temperature and resuspended to a final density of  $1-1.2 \times 10^6$  cells/mL in tissue-grade phosphate buffered saline. This procedure typically yielded >95% viability of cells as determined by trypan blue exclusion and easyCyte flow cytometry. To monitor S. aureus toxicity, 20 µL of cells were incubated with 20 µL of bacterial supernatant and incubated for 12 min at 37°C. For the USA300 strains, supernatants were diluted to 30% of the original volume in TSB as these isolates were considerably more toxic than the single-patient isolates. Cell death was quantified using easyCyte flow cytometry using the Guava viability stain according to manufacturer's instructions. Experiments were done in triplicate, and error bars indicate the average  $\pm$  the 95% confidence interval of multiple independent experiments.

## GWAS

The identification of genetic variation in all the clinical isolates studied has previously been described [25,26] with the exception of the 36 bacteraemic USA300 isolates. These were sequenced in an identical manner to the others; namely, genomic DNA was extracted using the QIAamp DNA Mini Kit (Qiagen), and unique index-tagged libraries were generated. Whole-genome sequencing was carried out using the Illumina HiSeq2000 with 100-base paired-end reads. Paired-end reads were mapped against the core chromosome of the ST8 USA300 reference genome sequence FPR3757 (accession NC\_02952) [52]. SNPs and indels were identified as described previously [53]. ENA accession numbers are listed in S1 Table. We conducted a quantitative association study on a set of 134 USA300 isolates to identify SNPs that were significantly associated with toxicity, using the PLINK software package (http://pngu.mgh.harvard.edu/purcell/plink/) [54]. These and a description of the loci are listed in S2 Table.

#### Maximum Likelihood Tree

This was estimated using PhyML with an HKY85 substitution model, empirical nucleotide usage, no rate heterogeneity, and no invariant sites. The percentage toxicity range was divided by three, where the most toxic isolates were labelled red, the midtoxicity isolates labelled orange, and the least toxic isolates labelled green.

## Endothelial Invasion Assay

Bacterial invasion of EA. Hy926 endothelial cells were performed as described previously with minor modifications [55]. Endothelial cells were cultured in Dulbecco's modified Eagles' medium (DMEM) supplemented with 10% FBS and 2 mM L-glutamine at 37°C in a humidified incubator with 5% CO<sub>2</sub>. Cells were liberated using trypsin-EDTA solution, resuspended in culture medium, and aliquoted into 24-well tissue culture tissue plates and grown to >95% confluence. Cells were washed twice in tissue-grade PBS, and 450 µL of fresh DMEM was added. 50 µL of washed *S. aureus* (1 x 10<sup>7</sup> CFU/mL) was added to the wells and incubated for 1 h at 37°C. Following incubation, the medium was aspirated and wells gently washed once in PBS and replaced with DMEM supplemented with 200 µg/mL gentamicin and incubated at 37°C for a further 60 min. Cells were subsequently lysed by the addition of 500 µL of Triton X-100, and bacterial CFU were enumerated by serial dilution of endothelial cell lysates and plating onto TSA plates and incubated at 37°C overnight. Experiments were performed in duplicate three times, and the error represents the 95% confidence interval.

## Formation of Neutrophil Extracellular Traps and Neutrophil Lysis Assay

Whole blood was obtained from healthy volunteers; ethical permission for all donations was obtained from a local research ethics committee (School of Pharmacy & Pharmaceutical Sciences Ethics Committee), and all participants gave written informed consent. Human neutrophils were isolated as previously described [56]. Cell-free supernatants from bacterial culture were diluted to 30% in warm Krebs buffer and then diluted 1:1 with prewarmed neutrophils ( $10^6$  neutrophils/mL in Krebs) and incubated at 37°C for 12 min. Cells were pelleted, and NET formation was quantified by measuring DNA content in the supernatant with Sytox Green and a DNA standard curve. Any signal from the bacterial culture was measured and subtracted from these values. For the neutrophil lysis assay, 20 µL of purified neutrophils was incubated with 20 µL of 10% bacterial supernatant for 15 min, and cell viability was assayed using Guava viability reagent and Guava flow cytometry.

## Protease Assay

A modified tryptic soy agar medium was made with 10% skim milk. 50  $\mu$ L of bacterial supernatant harvested from overnight cultivation was inoculated into 1 cm diameter wells perforated into the agar medium and incubated for 18 h at 37°C. The digested substrate, as a result of proteolytic activity, was observed as clear areas surrounding the wells, was measured. Protease assay were done in duplicate, three times, and error represents the 95% confidence interval.

## Antimicrobial Peptide Resistance

Purified human neutrophil defensin-1 (hNP-1) was purchased from AnaSpec Incorporated (California, USA). The hNP-1 susceptibility assay was performed in 1% BHI with the addition of 10 mM potassium phosphate buffer as described previously [57]. A final inoculum of  $10^5$  CFU, with a peptide concentration of 5 µg/mL, was employed and incubated for 2 h at 37°C. Final bacterial concentration was evaluated by serial plating onto TSA plates and data represented as mean (± SD) percent survival CFU/mL.

#### **Biofilm Formation**

Semiquantitative measurements of biofilm formation on 96-well polystyrene plates was determined based on the method of Ziebuhr et al [58]. Overnight bacteria grown in TSB were diluted 1:40 into 100  $\mu$ L TSB containing 1% glucose and grown for 24 h at 37°C. Following 24-h growth, plates were washed vigorously five times in PBS, dried and stained with 150  $\mu$ L of 1% crystal violet for 30 min at room temperature. Following five washes of PBS, wells were resuspended in 200  $\mu$ L of 7% acetic acid, and optical density at 595 nm was recorded using a plate reader.

#### **Relative Fitness Assay**

S. *aureus* isolates were grown overnight in BHI broth to an  $OD_{600}$  of 2.0 to ensure that all cells are in a similar physiological state at the start of the experiment. Competitions were established in TSB with and without 5% (vol/vol) freshly drawn human serum. The competition medium was inoculated with 10<sup>4</sup> CFU/mL of the marker strain (MSSA466, which is tetracycline resistant) and 10<sup>3</sup> CFU/mL of the test strain. Initial cell numbers were confirmed by plating. The bacteria were competed at 37°C in a shaking incubator (180 rpm) for 24 h. Final cell numbers were enumerated by serial dilutions on TSA plates (total cell count) and TSA plates containing 2 µg/mL tetracycline (marker strain count). The fitness of a strain was defined as a measure of the reproductive success of the population, which can be expressed as the natural logarithm of the ratio of the final and initial cell densities of the culture [59]. Each clinical strain was assayed once as each was considered a biological replicate indicative of its group.

relative (Darwinian) fitness  $= \frac{\ln(A(1)/A(0))}{\ln(M(1)/M(0))}$ 

where: A(0), estimated density of test strain at time 0; M(0), estimated density of marker strain at time 0; A(1), estimated density of test strain at time 1 d; M(1), estimated density of marker strain at time 1 d; ln, natural logarithm (logarithm to the base *e*).

#### Mathematical Model

We used a simple transmission model to examine the qualitative effect of toxicity-dependent probabilities of *S. aureus* to cause invasive disease (see S4 Fig for a flow diagram). We considered two different strains, distinguished by their level of toxicity (high, *h*, and low, *l*), and assumed that susceptible individuals, *S*, become colonised with strain *i* (I = h, l) upon contact with either colonised or infected individuals ( $C_i$  and  $I_i$ ) with transmission rates  $\beta_c$  and  $\beta_i$ , respectively. Individuals transition from colonisation to infection (i.e., SSTI) at a rate  $\delta_i$ , from which they either recover (at rate  $\tau_i$  and  $\rho_i$ , accounting for both treatment and immune mediated clearance) or go on to develop invasive diseases (bacteraemia,  $B_i$ ), with probability  $\sigma_i$ , which we assume does not contribute to transmission. The model can then be given by the following set of differential equations for the proportions of the population being susceptible, colonised, infected, or suffering from invasive disease:

$$\frac{dS}{dt} = \mu - S(\lambda_l + \lambda_h) + \tau_l I_l + \tau_h I_h - \mu S$$
$$\frac{dC_h}{dt} = \lambda_h S + \varrho_h I_h - (\mu + \delta_h) C_h - \nu C_h$$
$$\frac{dC_l}{dt} = \lambda_l S + \varrho_l I_l - (\mu + \delta_l) C_l + \nu C_h$$

$$\begin{split} \frac{dI_h}{dt} &= \delta_h C_h - \varrho_h I_h - (\mu + \sigma_h + \tau_h) I_h - \nu I_h \\ \frac{dI_l}{dt} &= \delta_l C_l - \varrho_l I_l - (\mu + \sigma_l + \tau_l) I_l + \nu I_h \\ \frac{dB_i}{dt} &= \sigma_i I_i - (\mu + \chi) B_i, \quad i = h, l \end{split}$$

with the force of infection of strain *i*,  $\lambda_i = \beta_C C_i + \beta_i I_i$ , disease-induced mortality  $\chi$ , and  $\mu$  as the natural birth/death rate. For simplicity, we did not allow for co- or superinfections but considered within-host evolution whereby more toxic strains can mutate (at a rate  $\nu$ ) towards lower levels of toxicity. Within this system, the probability of colonised individuals to develop infections (SSTI's) was assumed to be positively correlated with the strain's degree of toxicity (with  $\delta_h > \delta_l$ ), as were the transmission and treatment rates of infected individuals (i.e.,  $\beta_h > \beta_l$ ,  $\tau_h > \tau_l$ ). For illustration purposes only, we assumed that when the strains have equal probabilities to develop invasive disease (i.e.,  $\sigma_l = \sigma_2$ ), the less toxic strain has a higher fitness than the more toxic one. That is, we assumed that the less toxic strain is at an optimum, whereby toxicity-driven increases in transmissibility would be offset by higher clearance rates. Unless stated otherwise, we used the following parameter values:  $\mu = 0.017$ ,  $\beta_C = 0.05$ ,  $\beta_l = 4$ ,  $\beta_h = 4.4$ ,  $\delta_l = 2$ ,  $\delta_h = 2.2$ ,  $\tau_l = 3$ ,  $\tau_h = 3.3$ ,  $\rho_l = \rho_h = 10$ ,  $\nu = 0.002$ ,  $\chi = 5$ .

#### **Supporting Information**

S1 Data. A file containing three spreadsheets with the toxicity data for the clinical isolates used in this study, illustrated in <u>Fig 1</u>. (XLSX)

S2 Data. A spreadsheet containing the toxicity data for the transposon mutants used in this study, illustrated in <u>Fig 2</u>.

(XLSX)

S3 Data. An excel file containing five spreadsheets with the cell invasion, NETs, biofilm, hNP1 resistance, and protease activity for subset of isolates, illustrated in <u>Fig 4</u>. (XLSX)

S4 Data. A spreadsheet containing the relative fitness data for the isolates tested here with and without human serum, illustrated in <u>Fig 5</u>. (XLSX)

S5 Data. An excel file containing two spreadsheets with the toxicity data for a range of *S. aureus* toxin mutants and purified toxins; and comparing toxicity using T2 cells and fresh human neutrophils, illustrated in <u>S1 Fig</u>. (XLSX)

S6 Data. A spreadsheet containing growth dynamic data for high- and low-toxicity isolates with and without human serum, illustrated in <u>S2 Fig</u>. (XLSX)

**S1 Fig. T2 cell susceptibility to** *S. aureus* **cytolytic toxins.** A: Using a combination of isogenic mutant strains and purified toxins, the susceptibility of the T2 cell line is illustrated. Mean of six replicates are presented, error bars represent the 95% confidence intervals, \* indicates

statistically significant differences. B: The toxicity to both T2 cells and neutrophils of subset of six high- and six low-toxicity isolates from the single-patient collection was quantified. The effect of the low-toxicity isolates on cell death does not vary when fresh human neutrophils that are susceptible to LukAB and LukED are used. To access this data, see <u>S5 Data</u>. (TIF)

**S2 Fig. Growth dynamics of 10 high- and 10 low-toxicity isolates in BHI with and without 5% (vol/vol) human serum.** To access this data, see <u>S6 Data</u>. (TIF)

**S3 Fig. Toxicity- and bacteraemia-dependent fitness landscape.** A strain's fitness, here shown as its force of infection at equilibrium, is determined by its level of toxicity and its propensity to cause bacteraemia. Due to an evolutionary trade-off between toxicity-driven increase in transmissibility and treatment rate, fitness initially increases with enhanced toxicity but then declines as individuals become more likely to seek treatment faster, thus limiting the opportunity for onward transmission. With equal probabilities to cause bacteraemia (scenario 1), a more toxic strain (red) can therefore be outcompeted by a strain with lower levels of toxicity (green). In contrast, by assuming an inverse correlation between toxicity and the probability of causing bacteraemia (scenario 2), the more toxic strain can gain a competitive advantage, leading to the exclusion of the less toxic strain. (TIF)

S4 Fig. Flow diagram illustrating the dynamical processes underlying the mathematical model. The population is subdivided into different classes representing those susceptible (*S*), colonised with strain i (C\_i), infected (SSTI) with strain i ( $I_i$ ), and those bacteraemic ( $B_i$ ). (TIF)

**S1 Table.** *S. aureus* isolates used in the study. (DOCX)

**S2 Table. List of SNPs associated with changes in toxicity as determined by PLINK.** SNP site: number corresponds to the distance from the origin of replication. Gene: gene in which the toxicity-associated SNP resides. Description: known activity of the gene. AA change: amino acid change caused by the SNP. Tn mutant: name of the Tn mutant obtained from the Nebraska library in the relevant gene. Bold font indicates Tn mutants with functionally verified effect on toxicity.

(DOCX)

**S3** Table. Clinical features of the bacteraemic isolates used in this study. (DOCX)

## Acknowledgments

We would like to thank Prof. Laurence Hurst for helpful comments during the preparation of this manuscript and Beth Blane for technical assistance. Stephen R. Clarke was unable to confirm authorship, so Ruth Massey, the corresponding author, has confirmed authorship on his behalf.

## **Author Contributions**

Conceived and designed the experiments: ML ACU FDL SJP SRC DJW MR RCM. Performed the experiments: ML ACU EDA MY KO HAT BW MP SRC JD MH AAV RB BCY DJW MR

RCM. Analyzed the data: ML DJW MR RCM. Contributed reagents/materials/analysis tools: ACU FDL EF SJP DWC BCY DJW RCM. Wrote the paper: ML DJW MR RCM.

#### References

- 1. Welch RA (1991) Pore-forming cytolysins of gram-negative bacteria. Mol Microbiol 5: 521–528. PMID: 2046545
- Sears CL, Kaper JB (1996) Enteric bacterial toxins: mechanisms of action and linkage to intestinal secretion. Microbiol Rev 60: 167–215. PMID: <u>8852900</u>
- Dinges MM, Orwin PM, Schlievert PM (2000) Exotoxins of Staphylococcus aureus. Clin Microbiol Rev 13: 16–34, table of contents. PMID: <u>10627489</u>
- Lyras D, O'Connor JR, Howarth PM, Sambol SP, Carter GP, et al. (2009) Toxin B is essential for virulence of Clostridium difficile. Nature 458: 1176–1179. doi: 10.1038/nature07822 PMID: 19252482
- 5. Otto M (2010) Basis of virulence in community-associated methicillin-resistant Staphylococcus aureus. Annu Rev Microbiol 64: 143–162. doi: <u>10.1146/annurev.micro.112408.134309</u> PMID: <u>20825344</u>
- 6. Foster TJ (2005) Immune evasion by staphylococci. Nat Rev Microbiol 3: 948–958. PMID: 16322743
- Rooijakkers SH, van Kessel KP, van Strijp JA (2005) Staphylococcal innate immune evasion. Trends Microbiol 13: 596–601. PMID: <u>16242332</u>
- Gordon RJ, Lowy FD (2008) Pathogenesis of methicillin-resistant Staphylococcus aureus infection. Clin Infect Dis 46 Suppl 5: S350–359. doi: 10.1086/533591 PMID: 18462090
- 9. Lowy FD (1998) Staphylococcus aureus infections. N Engl J Med 339: 520–532. PMID: 9709046
- Somerville GA, Beres SB, Fitzgerald JR, DeLeo FR, Cole RL, et al. (2002) In vitro serial passage of Staphylococcus aureus: changes in physiology, virulence factor production, and agr nucleotide sequence. J Bacteriol 184: 1430–1437. PMID: <u>11844774</u>
- Sandoz KM, Mitzimberg SM, Schuster M (2007) Social cheating in Pseudomonas aeruginosa quorum sensing. Proc Natl Acad Sci U S A 104: 15876–15881. PMID: <u>17898171</u>
- Cheung GY, Kretschmer D, Duong AC, Yeh AJ, Ho TV, et al. (2014) Production of an attenuated phenol-soluble modulin variant unique to the MRSA clonal complex 30 increases severity of bloodstream infection. PLoS Pathog 10: e1004298. doi: <u>10.1371/journal.ppat.1004298</u> PMID: <u>25144687</u>
- Rose HR, Holzman RS, Altman DR, Smyth DS, Wasserman GA, et al. (2014) Cytotoxic Virulence Predicts Mortality in Nosocomial Pneumonia Due to Methicillin-Resistant Staphylococcus aureus. J Infect Dis 2015 211(12):1862–74. doi: 10.1093/infdis/jiu554 PMID: 25298028
- Smyth DS, Kafer JM, Wasserman GA, Velickovic L, Mathema B, et al. (2012) Nasal carriage as a source of agr-defective Staphylococcus aureus bacteremia. J Infect Dis 206: 1168–1177. PMID: 22859823
- Fowler VG Jr., Sakoulas G, McIntyre LM, Meka VG, Arbeit RD, et al. (2004) Persistent bacteremia due to methicillin-resistant Staphylococcus aureus infection is associated with agr dysfunction and lowlevel in vitro resistance to thrombin-induced platelet microbicidal protein. J Infect Dis 190: 1140–1149. PMID: <u>15319865</u>
- Traber KE, Lee E, Benson S, Corrigan R, Cantera M, et al. (2008) agr function in clinical Staphylococcus aureus isolates. Microbiology 154: 2265–2274. doi: <u>10.1099/mic.0.2007/011874-0</u> PMID: <u>18667559</u>
- Shopsin B, Eaton C, Wasserman GA, Mathema B, Adhikari RP, et al. (2010) Mutations in agr do not persist in natural populations of methicillin-resistant Staphylococcus aureus. J Infect Dis 202: 1593– 1599. doi: 10.1086/656915 PMID: 20942648
- Diep BA, Otto M (2008) The role of virulence determinants in community-associated MRSA pathogenesis. Trends Microbiol 16: 361–369. doi: <u>10.1016/j.tim.2008.05.002</u> PMID: <u>18585915</u>
- Chua KY, Monk IR, Lin YH, Seemann T, Tuck KL, et al. (2014) Hyperexpression of alpha-hemolysin explains enhanced virulence of sequence type 93 community-associated methicillin-resistant Staphylococcus aureus. BMC Microbiol 14: 31. doi: <u>10.1186/1471-2180-14-31</u> PMID: <u>24512075</u>
- Heyer G, Saba S, Adamo R, Rush W, Soong G, et al. (2002) Staphylococcus aureus agr and sarA functions are required for invasive infection but not inflammatory responses in the lung. Infect Immun 70: 127–133. PMID: <u>11748173</u>
- Gillaspy AF, Hickmon SG, Skinner RA, Thomas JR, Nelson CL, et al. (1995) Role of the accessory gene regulator (agr) in pathogenesis of staphylococcal osteomyelitis. Infect Immun 63: 3373–3380. PMID: <u>7642265</u>

- Cheung AL, Eberhardt KJ, Chung E, Yeaman MR, Sullam PM, et al. (1994) Diminished virulence of a sar-/agr- mutant of Staphylococcus aureus in the rabbit model of endocarditis. J Clin Invest 94: 1815– 1822. PMID: <u>7962526</u>
- Abdelnour A, Arvidson S, Bremell T, Ryden C, Tarkowski A (1993) The accessory gene regulator (agr) controls Staphylococcus aureus virulence in a murine arthritis model. Infect Immun 61: 3879–3885. PMID: 8359909
- Rudkin JK, Edwards AM, Bowden MG, Brown EL, Pozzi C, et al. (2012) Methicillin resistance reduces the virulence of healthcare-associated methicillin-resistant Staphylococcus aureus by interfering with the agr quorum sensing system. J Infect Dis 205: 798–806. doi: <u>10.1093/infdis/jir845</u> PMID: <u>22301683</u>
- Young BC, Golubchik T, Batty EM, Fung R, Larner-Svensson H, et al. (2012) Evolutionary dynamics of Staphylococcus aureus during progression from carriage to disease. Proc Natl Acad Sci U S A 109: 4550–4555. doi: 10.1073/pnas.1113219109 PMID: 22393007
- 26. Uhlemann AC, Dordel J, Knox JR, Raven KE, Parkhill J, et al. (2014) Molecular tracing of the emergence, diversification, and transmission of S. aureus sequence type 8 in a New York community. Proc Natl Acad Sci U S A 111: 6738–6743. doi: 10.1073/pnas.1401006111 PMID: 24753569
- Li R, Coulthard LG, Wu MC, Taylor SM, Woodruff TM (2013) C5L2: a controversial receptor of complement anaphylatoxin, C5a. FASEB J 27: 855–864. doi: <u>10.1096/fj.12-220509</u> PMID: <u>23239822</u>
- Lei MG, Cue D, Roux CM, Dunman PM, Lee CY (2011) Rsp inhibits attachment and biofilm formation by repressing fnbA in Staphylococcus aureus MW2. J Bacteriol 193: 5231–5241. doi: <u>10.1128/JB.</u> <u>05454-11</u> PMID: <u>21804010</u>
- Fey PD, Endres JL, Yajjala VK, Widhelm TJ, Boissy RJ, et al. (2013) A genetic resource for rapid and comprehensive phenotype screening of nonessential Staphylococcus aureus genes. MBio 4: e00537– 00512. doi: 10.1128/mBio.00537-12 PMID: 23404398
- Laabei M, Recker M, Rudkin JK, Aldeljawi M, Gulay Z, et al. (2014) Predicting the virulence of MRSA from its genome sequence. Genome Res 24: 839–849. doi: <u>10.1101/gr.165415.113</u> PMID: <u>24717264</u>
- Luong TT, Sau K, Roux C, Sau S, Dunman PM, et al. (2011) Staphylococcus aureus ClpC divergently regulates capsule via sae and codY in strain newman but activates capsule via codY in strain UAMS-1 and in strain Newman with repaired saeS. J Bacteriol 193: 686–694. doi: <u>10.1128/JB.00987-10</u> PMID: <u>21131496</u>
- Somerville GA, Proctor RA (2009) At the crossroads of bacterial metabolism and virulence factor synthesis in Staphylococci. Microbiol Mol Biol Rev 73: 233–248. doi: <u>10.1128/MMBR.00005-09</u> PMID: <u>19487727</u>
- Casadevall A, Pirofski LA (2003) The damage-response framework of microbial pathogenesis. Nat Rev Microbiol 1: 17–24. PMID: <u>15040176</u>
- Arvidson S, Tegmark K (2001) Regulation of virulence determinants in Staphylococcus aureus. Int J Med Microbiol 291: 159–170. PMID: <u>11437338</u>
- Bronner S, Monteil H, Prevost G (2004) Regulation of virulence determinants in Staphylococcus aureus: complexity and applications. FEMS Microbiol Rev 28: 183–200. PMID: <u>15109784</u>
- Pilsczek FH, Salina D, Poon KK, Fahey C, Yipp BG, et al. (2010) A novel mechanism of rapid nuclear neutrophil extracellular trap formation in response to Staphylococcus aureus. J Immunol 185: 7413– 7425. doi: <u>10.4049/jimmunol.1000675</u> PMID: <u>21098229</u>
- Oogai Y, Matsuo M, Hashimoto M, Kato F, Sugai M, et al. (2011) Expression of virulence factors by Staphylococcus aureus grown in serum. Appl Environ Microbiol 77: 8097–8105. doi: <u>10.1128/AEM.</u> <u>05316-11</u> PMID: <u>21926198</u>
- Malachowa N, Whitney AR, Kobayashi SD, Sturdevant DE, Kennedy AD, et al. (2011) Global changes in Staphylococcus aureus gene expression in human blood. PLoS One 6: e18617. doi: <u>10.1371/</u> journal.pone.0018617 PMID: 21525981
- Surewaard BG, Nijland R, Spaan AN, Kruijtzer JA, de Haas CJ, et al. (2012) Inactivation of staphylococcal phenol soluble modulins by serum lipoprotein particles. PLoS Pathog 8: e1002606. doi: <u>10.</u> 1371/journal.ppat.1002606 PMID: 22457627
- Kaasch AJ, Barlow G, Edgeworth JD, Fowler VG Jr., Hellmich M, et al. (2014) Staphylococcus aureus bloodstream infection: a pooled analysis of five prospective, observational studies. J Infect 68: 242– 251. doi: 10.1016/j.jinf.2013.10.015 PMID: 24247070
- Koser CU, Holden MT, Ellington MJ, Cartwright EJ, Brown NM, et al. (2012) Rapid whole-genome sequencing for investigation of a neonatal MRSA outbreak. N Engl J Med 366: 2267–2275. doi: <u>10.</u> <u>1056/NEJMoa1109910</u> PMID: <u>22693998</u>
- Sherry NL, Porter JL, Seemann T, Watkins A, Stinear TP, et al. (2013) Outbreak investigation using high-throughput genome sequencing within a diagnostic microbiology laboratory. J Clin Microbiol 51: 1396–1401. doi: <u>10.1128/JCM.03332-12</u> PMID: <u>23408689</u>

- Castillo-Ramirez S, Harris SR, Holden MT, He M, Parkhill J, et al. (2011) The impact of recombination on dN/dS within recently emerged bacterial clones. PLoS Pathog 7: e1002129. doi: <u>10.1371/journal.</u> <u>ppat.1002129</u> PMID: <u>21779170</u>
- Castillo-Ramirez S, Corander J, Marttinen P, Aldeljawi M, Hanage WP, et al. (2012) Phylogeographic variation in recombination rates within a global clone of methicillin-resistant Staphylococcus aureus. Genome Biol 13: R126. doi: <u>10.1186/gb-2012-13-12-r126</u> PMID: <u>23270620</u>
- 45. Sheppard SK, Didelot X, Meric G, Torralbo A, Jolley KA, et al. (2013) Genome-wide association study identifies vitamin B5 biosynthesis as a host specificity factor in Campylobacter. Proc Natl Acad Sci U S A 110: 11923–11927. doi: 10.1073/pnas.1305559110 PMID: 23818615
- 46. Farhat MR, Shapiro BJ, Kieser KJ, Sultana R, Jacobson KR, et al. (2013) Genomic analysis identifies targets of convergent positive selection in drug-resistant Mycobacterium tuberculosis. Nat Genet 45: 1183–1189. doi: <u>10.1038/ng.2747</u> PMID: <u>23995135</u>
- Marvig RL, Sommer LM, Molin S, Johansen HK (2015) Convergent evolution and adaptation of Pseudomonas aeruginosa within patients with cystic fibrosis. Nat Genet 47: 57–64. doi: <u>10.1038/ng.3148</u> PMID: <u>25401299</u>
- Read TD, Massey RC (2014) Characterizing the genetic basis of bacterial phenotypes using genomewide association studies: a new direction for bacteriology. Genome Med 6: 109. doi: <u>10.1186/s13073-</u> 014-0109-z PMID: 25593593
- Alam MT, Petit RA 3rd, Crispell EK, Thornton TA, Conneely KN, et al. (2014) Dissecting vancomycinintermediate resistance in staphylococcus aureus using genome-wide association. Genome Biol Evol 6: 1174–1185. doi: <u>10.1093/gbe/evu092</u> PMID: <u>24787619</u>
- McVicker G, Prajsnar TK, Williams A, Wagner NL, Boots M, et al. (2014) Clonal expansion during Staphylococcus aureus infection dynamics reveals the effect of antibiotic intervention. PLoS Pathog 10: e1003959. doi: 10.1371/journal.ppat.1003959 PMID: 24586163
- Collins J, Buckling A, Massey RC (2008) Identification of factors contributing to T-cell toxicity of Staphylococcus aureus clinical isolates. J Clin Microbiol 46: 2112–2114. doi: <u>10.1128/JCM.00156-08</u> PMID: <u>18417658</u>
- Diep BA, Gill SR, Chang RF, Phan TH, Chen JH, et al. (2006) Complete genome sequence of USA300, an epidemic clone of community-acquired meticillin-resistant Staphylococcus aureus. Lancet 367: 731–739. PMID: 16517273
- Croucher NJ, Harris SR, Fraser C, Quail MA, Burton J, et al. (2011) Rapid pneumococcal evolution in response to clinical interventions. Science 331: 430–434. doi: <u>10.1126/science.1198545</u> PMID: <u>21273480</u>
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MA, et al. (2007) PLINK: a tool set for wholegenome association and population-based linkage analyses. Am J Hum Genet 81: 559–575. PMID: 17701901
- 55. Edwards AM, Potts JR, Josefsson E, Massey RC (2010) Staphylococcus aureus host cell invasion and virulence in sepsis is facilitated by the multiple repeats within FnBPA. PLoS Pathog 6: e1000964. doi: 10.1371/journal.ppat.1000964 PMID: 20585570
- 56. Thomas CP, Clark SR, Hammond VJ, Aldrovandi M, Collins PW, et al. (2014) Identification and quantification of aminophospholipid molecular species on the surface of apoptotic and activated cells. Nat Protoc 9: 51–63. doi: 10.1038/nprot.2013.163 PMID: 24336470
- 57. Mishra NN, Yang SJ, Chen L, Muller C, Saleh-Mghir A, et al. (2013) Emergence of daptomycin resistance in daptomycin-naive rabbits with methicillin-resistant Staphylococcus aureus prosthetic joint infection is associated with resistance to host defense cationic peptides and mprF polymorphisms. PLoS One 8: e71151. doi: 10.1371/journal.pone.0071151 PMID: 23990934
- Ziebuhr W, Heilmann C, Gotz F, Meyer P, Wilms K, et al. (1997) Detection of the intercellular adhesion gene cluster (ica) and phase variation in Staphylococcus epidermidis blood culture strains and mucosal isolates. Infect Immun 65: 890–896. PMID: 9038293
- Lenski RE, Rose MR, Simpson SC, Tadler SC (1991) Long-term experimental evolution in Escherichia coli. I. Adaptation and divergence during 2,000 generations. The American Naturalist 138: 26.